Author's response to reviews

Title: Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complex: a prospective cohort study

Authors:

Rajpal S. Kashyap (raj_ciims@rediffmail.com)
Anju N Rajan (anju_mudaliar@rediffmail.com)
Sonali S Ramteke (sonaliramteke8@yahoo.co.in)
Vijay S Agrawal (ciims_ngp@sancharnet.in)
Sanjivani S Kelkar (ciims_ngp@sancharnet.in)
Hemant J Purohit (hemantdrd@hotmail.com)
Girdhar M Taori (taorigm_ciims@yahoo.co.in)
Hatim F Darginawala (hfd_ciims@rediffmail.com)

Version: 6 Date: 9 July 2007

Author's response to reviews: see over
To,

The BioMed Central Editorial Production Team

Authors: Rajpal S. Kashyap, Anju N Rajan, Sonali S Ramteke, Vijay S Agrawal, Sanjivani S Kelkar, Hemant J Purohit, Girdhar M Taori and Hatim F Daginawala
Title: Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complex: a prospective cohort study
Journal: BMC Infectious Diseases
MS: 3516515591255519

Dear Sir,

Please find enclosed revised copy of the manuscript (MS: 3516515591255519). Format is changed as per the comments.

Please let me know if anything needed from my side.

Thanks for your cooperation.

With kind regards